Author's response to reviews

Title: Intramural haematoma of the thoracic aorta: who is to be alerted the cardiologist or the cardiac surgeon?

Authors:

NIKOLAOS G. BAIKOSSIS (ngbaik@yahoo.com)
EFSTRATIOS E. APOSTOLAKIS (stratisapostolakis@yahoo.gr)
STAVROS N. SIMINELAKIS (ssiminel@yahoo.gr)
GEORGIOS S. PAPADOPoulos (baikoussis@yahoocom)
JOHN GOUDEVENOS (ngbaik@yahoo.com)

Version: 3 Date: 4 September 2009

Author's response to reviews: see over
Dear Sir/ Madame

We herewith submit for publication in *j of cardiothoracic surgery* our review paper entitled: **Intramural Haematoma of the Thoracic Aorta: an emergency for the cardiologist or for the Cardiac Surgeon?**

**Nikolaos G. Baikoussis MD, Efstratios E. Apostolakis MD, PhD, Stavros N. Siminelakis MD, PhD, Georgios S. Papadopoulos MD, PhD, John Goudevenos MD, PhD.**

The paper is prepared following the instructions of your journal on line. The author indicates that this manuscript should be considered and published as review article.

*We correct the English grammar and language according your instruction.*

We corrected the manuscript according the reviewers comments.

Reviewer 1: We added the definition in the introduction.
We used the article by Task Force as proposed and the article of Chao et al. Minor corrections are also done. The majority of his suggestions is acceptable and performed.

Reviewer 2: He did not propose any correction. He found the manuscript interesting.

Reviewer 3: Recent studies are added into the manuscript. The reviewer is right. Some minor correction are also done.

Reviewer 4: Unfortunately we could not find specific data about correlation between atherosclerotic aorta and IMH. The other requested details are also not provided.

Sincerely yours

Nikolaos G. Baikoussis MD